Author’s response to reviews

Title: Attentional bias modification in reducing test anxiety vulnerability: A randomized controlled trial

Authors:

Wenpeng Cai (wpcai@smmu.edu.cn)
Yu Pan (531828485@qq.com)
Huangyangzi Chai (yokochyz@163.com)
Yi Cui (cuiyi210@163.com)
Jin Yan (yanjingk@qq.com)
Wei Dong (sophiedongwei@163.com)
Guanghui Deng (bfbedu@126.com)

Version: 2 Date: 22 Jun 2017

Author’s response to reviews:

Trial registration information:

Thank you for providing the trial registration number (TRN) for your trial. Please can you clarify the registry in which your trial has been registered?

Please include the trial registration number (TRN) and date of registration as the last line of the manuscript abstract. Please list your trial registry, along with the unique identifying number, the date of registration and indicate the retrospective nature of registration (e.g. Trial registration: Current Controlled Trials ISRCTN73824458. Registered 28 September 2004). Please note that there should be no space between the letters and numbers of your trial registration number.

Response: “This trial was retrospectively registered on June 22, 2017 with the registration number ChiCTR-IOR-17011745 and the title ‘Attentional Bias in high anxiety individuals and its modification’.” This statement was added in the abstract.